Author’s response to reviews

Title: Leiomyosarcoma with partial rhabdomyoblastic differentiation: First case report of primary cardiac origin

Authors:

Yoichiro Okubo (yoichiro0207@med.toho-u.ac.jp)
Kazutoshi Shibuya (kaz@med.toho-u.ac.jp)
Atsushi Namiki (namiki@med.toho-u.ac.jp)
Kazuhisa Takamura (kazuhisa.t@med.toho-u.ac.jp)
Noriaki Kameda (nkameda@med.toho-u.ac.jp)
Tetsuo Nemoto (tetsuo.nemoto@med.toho-u.ac.jp)
Aki Mitsuda (akimitsuda@med.toho-u.ac.jp)
Megumi Wakayama (megumi.wakayama@med.toho-u.ac.jp)
Minoru Shinozaki (ushino@med.toho-u.ac.jp)
Nobuyuki Hiruta (nhir@med.toho-u.ac.jp)
Kanako Kitahara (kit@med.toho-u.ac.jp)
Takao Ishiwatari (takaboo1982@yahoo.co.jp)
Junichi Yamazaki (yamasaki@med.toho-u.ac.jp)

Version: 3 Date: 9 February 2011

Author’s response to reviews: see over
The editorial office of the BMC cancer

Dear the BMC Series Editorial Production Team

We sincerely appreciate your kindest comments.

We have carefully revised the manuscript with following to Editorial Production Team’s comments,

and are sending herewith a revised manuscript.

Yours sincerely,

First author: Yoichiro Okubo, E-mail: yoichiro0207@med.toho-u.ac.jp

Corresponding author: Kazuhisa Takamura, E-mail: kazuhisa.t@med.toho-u.ac.jp

Division of Cardiovascular Medicine, Department of Internal Medicine, Toho University Omori Medical Center, Toho University School of Medicine, 6-11-1 Omori-Nishi, Ota-Ku, Tokyo, 143-8541, Japan

Phone: +81-3-3762-4151, Fax: +81-3-3766-7810
Responses for Editorial Production Team’s comments

Major revisions

Comment 1

[Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution]

Our response: We revised and added affiliations to the requested format.

Comment 2

[Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.]

Our response: Thank you for your recommendation. We added “tables section” following the references section to be visible within the final published manuscript.
Minor revisions

Comment 1

[Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.]

Our response: We removed all highlighting text from our manuscript.

Comment 2

[Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.]

Our response: We checked typographical and other errors in our manuscript carefully again.